Dear editor,

I have finished the revision of the manuscript “Synchronous renal pelvis carcinoma associated with small lymphocytic lymphoma: a case report” (Manuscript NO.: 77245, Case Report) according to all reviewers’ comments. I hope the revised manuscript can be accepted for publishing. The following content is a point–by point response.

**Reviewer 1#:**

a) It must be structured in Introduction, Methods and materials, Results, Discussion and Conclusions.

**Response:**

It has been structured according to the above request.

b) In the introduction it is necessary to review the state of the art. And for this, a representative bibliography must be sought.

**Response:**

Only five cases of coincident CLL/SLL and bladder cancer have been reported. No coincident CLL/SLL and UTUCs case has been reported in the literature.

c) In the discussion, the presented work should be compared with other similar works, establishing what the limitations and advances are.

**Response:**


d) In the conclusions, the scientific contribution of the article must be clearly established.

**Response:**


**Reviewer 2#:**

Good file design and good language expression but apart from the good language command and good work design what is new that the paper present it is a mere coincidence no more no less I hope this may. Not agitate the authors nor thre. Editors

**Response:**
No need for revision.

**Science editor comments:**

Language Quality: Grade B (Minor language polishing)

Scientific Quality: Grade B (Very good)

Language polishing finished

**Company editor-in-chief comments:**

Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. The author needs to add the following copyright information to the bottom right-hand side of the picture in PowerPoint (PPT): Copyright ©The Author(s) 2022, and submit as “77245-Figures.pptx”. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

**Response:**

The manuscript is submitted according to the above requests.